Author's response to reviews

Title: PKC alpha regulates netrin-1/UNC5B-mediated survival pathway in bladder cancer.

Authors:

Jiao Liu (546439049@qq.com)  
Chui ze Kong (494222088@qq.com)  
Da xin Gong (sylujiiao@sohu.com)  
Zhe zhang (cm_xd@sohu.com)  
Yu yan Zhu (15840259855@163.com)

Version: 2  
Date: 19 November 2013

Author's response to reviews:

I have changed all comments which were suggested by reviewers and editors, including abstract, introduction, methods, results, discussion, figure1 & 2 and table1 & 2.